International Journal of Linguistics, Literature, and Culture



REVIEW HISTORY

Paper: "Migration and Religion: A Case of Healing Arabic Amulets among Syriac Orthodox Immigrants in Canada"

Corresponding Author: Guita Hourani

Email: ghourani@ndu.edu.lb

Doi: 10.19044/Ilc.v6no3a3

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 30.09.2019

